Author’s response to reviews

Title: Iodine level concentration, coverage of adequately iodized salt consumption and factors affecting proper iodized salt utilization among households in North Ethiopia: A Community Based Cross sectional Study

Authors:

Abraham Desta (abaregaydesta@gmail.com)
Usha Kulkarni (ushakulkarni@hotmail.com)
Kidan Abrha (kidanabrha@yahoo.com)
Solomon Worku (solomonw@tutape.org)
Berhe Sahle (berheph@gmail.com)

Version: 2 Date: 24 Mar 2019

Author’s response to reviews:

24th March, 2019

Dear Sir/ Madam

Editor of BMC Nutrition

As per your suggestions

1. I have amended through paraphrasing on the specified lines of the previous version of the manuscripts to avoid text overlap with the previously published works as suggested by the reviewers and editors.

2. I have added the Consent for publication in the Declarations section

3. I have also amended the author’s contribution as per the guideline
4. I have incorporated the questionnaire as a supplementary material

5. Finally, I have cleaned all highlights and tracks from the final manuscript to submit the final version.

Thank you for your constructive comments!

Abraham Aregay Desta

Tigray Health Research Institute, Mekelle, Tigray, Ethiopia